Author’s response to reviews

Title: Association of an NF-kappaB1 intron SNP (rs4648068) with gastric cancer patients in Chinese Han population

Authors:

renquan lu (lurenquan66@hotmail.com)
Yin Chen (1251487447@qq.com)
Xiang gao (gxcy2001@hotmail.com)
Ran Xiao (ljerry19890505@hotmail.com)
Jian Ni (jiannihome@yahoo.com)
Sheng Li (lisheng1996@gmail.com)
Lin Guo (jykca@sina.com)

Version: 2 Date: 15 November 2011

Author’s response to reviews:

we have revised the title page and made the author list. And, we added the statement to the ethics in the Methods section of the manuscript and declared there was no competing interests. Finally, we edited the authors' contributions and the acknowledgements.